Author's response to reviews

Title: How are falls and fear of falling associated with objectively measured physical activity in a cohort of community-dwelling older men?

Authors:

Barbara J Jefferis (b.jefferis@ucl.ac.uk)
Steve Illiffe (s.illiffe@ucl.ac.uk)
Denise Kendrick (Denise.Kendrick@nottingham.ac.uk)
Ngaire Kerse (n.kerse@auckland.ac.nz)
Stewart Trost (s.trost@uq.edu.au)
Lucy T Lennon (l.lennon@ucl.ac.uk)
Sarah Ash (sarah.ash@ucl.ac.uk)
Claudio Sartini (c.sartini@ucl.ac.uk)
Richard W Morris (richard.morris@ucl.ac.uk)
Goya Wannamethee (g.wannamethee@ucl.ac.uk)
Peter H Whincup (pwhincup@sgul.ac.uk)

Version: 2 Date: 21 May 2014

Author's response to reviews:

I attach a revised version of the paper with the page numbers and line numbers now added, as described in the letter attached above for the "response to review"